Check for updates

OPEN LETTER

Considerations in planning a controlled human infection

model in at-risk groups in sub-Saharan Africa: the case for

pneumococcal challenge in people living with HIV in Malawi

and a report of stakeholder consultation

[version 1; peer review: awaiting peer review]

Klara Doherty^[]^{1,2*}, Anthony Chirwa^{2*}, Shalom Songolo², Alice Kusakala², Edna Nsomba^[]², Pemphero Liwonde², Daniela Ferreira^{1,3}, Henry Mwandumba^{1,2}, Kondwani Jambo^[]^{1,2}, Stephen Gordon^[]^{1,2}

¹Liverpool School of Tropical Medicine Department of Clinical Sciences, Liverpool, England, UK ²Malawi-Liverpool Wellcome programme, Blantyre, Malawi ³University of Oxford Oxford Vaccine Group, Oxford, England, UK

* Equal contributors

v1	t published: 07 Nov 2024, 9 :655 s://doi.org/10.12688/wellcomeopenres.23277.1	
	Latest published: 07 Nov 2024, 9 :655 https://doi.org/10.12688/wellcomeopenres.23277.1	

Abstract

Controlled human infection models offer a unique opportunity to understand infectious disease pathogenesis and have accelerated vaccine development and evaluations in malaria and typhoid. One major limitation of most CHIMs is that they are typically conducted in healthy young adults who are generally the population least affected by infectious disease, and who exhibit distinct disease profiles to more at-risk populations such as people living with HIV, young children, and older adults. However, the added value of studying these populations with high relevance is only desirable if it can be done safely, robustly and acceptably. We present a framework to guide the conduct of a controlled human infection model in people living with HIV using a case-example of an experimental human pneumococcal carriage model in a setting of high disease-burden and transmission.

Plain language summary

Controlled human infection models (CHIMs) are a research method in which an infection is safely introduced into volunteer participants to better in order to understand the infection and test vaccines against it. These models are more efficient than traditional clinical studies because they require fewer participants. Most CHIMs have focused on

Open Peer Review

Approval Status AWAITING PEER REVIEW

Any reports and responses or comments on the article can be found at the end of the article.

healthy young adults, who are not the ones most affected by infectious diseases and have a distinct infection profile and vaccine response compared to population with a greater burden of infectious disease such as older adults and people living with HIV. Recently researchers have started to include these high-burden populations in CHIMs but safety and ethics are critical considerations before embarking on such studies. We propose a framework for safely and ethically conducting CHIMs in people living with HIV in order to advance research in this key population. We use a case example of a CHIM of pneumococcal carriage in the nose of people living with HIV in Malawi.

Keywords

pneumococcal infections, HIV infections, vaccines, Malawi, Streptococcus pneumoniae

Corresponding author: Klara Doherty (kdoherty@mlw.mw)

Author roles: Doherty K: Conceptualization, Data Curation, Formal Analysis, Funding Acquisition, Investigation, Methodology, Project Administration, Resources, Software, Supervision, Visualization, Writing – Original Draft Preparation, Writing – Review & Editing; Chirwa A: Conceptualization, Data Curation, Formal Analysis, Funding Acquisition, Investigation, Methodology, Project Administration, Supervision, Writing – Original Draft Preparation, Writing – Review & Editing; Songolo S: Data Curation, Writing – Review & Editing; Kusakala A: Data Curation, Writing – Review & Editing; Nsomba E: Data Curation, Writing – Review & Editing; Liwonde P: Data Curation, Writing – Review & Editing; Ferreira D: Conceptualization, Supervision, Writing – Review & Editing; Mwandumba H: Supervision, Writing – Review & Editing; Jambo K: Conceptualization, Supervision, Writing – Review & Editing; Gordon S: Conceptualization, Funding Acquisition, Resources, Supervision, Writing – Review & Editing

Competing interests: No competing interests were disclosed.

Grant information: This work was supported by Wellcome [226731/Z/22/Z] and [211433/Z/18/Z]. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Copyright: © 2024 Doherty K *et al.* This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

How to cite this article: Doherty K, Chirwa A, Songolo S *et al.* Considerations in planning a controlled human infection model in atrisk groups in sub-Saharan Africa: the case for pneumococcal challenge in people living with HIV in Malawi and a report of stakeholder consultation [version 1; peer review: awaiting peer review] Wellcome Open Research 2024, **9**:655 https://doi.org/10.12688/wellcomeopenres.23277.1

First published: 07 Nov 2024, 9:655 https://doi.org/10.12688/wellcomeopenres.23277.1

Disclaimer

The views expressed in this article are those of the author(s). Publication in Wellcome Open Research does not imply endorsement by Wellcome

Introduction

Controlled human infection models (CHIMs) involve the deliberate introduction of an infectious agent to volunteer participants in order to study infectious disease pathogenesis or test a therapeutic intervention. In recent years CHIMs established in high-income, low endemic regions have been transferred to endemic settings to study the diseases in the populations most affected¹⁻⁴. A malaria CHIM established in Oxford and Maryland has been transferred to malaria endemic settings in Kenya, Tanzania, and Gabon, to study the disease in those with prior immunity and to make the findings relevant to those most affected⁴⁻⁶. Similarly, an experimental human pneumococcal carriage model (EHPC) established in Liverpool has been transferred to Malawi where post-vaccine persistence of pneumococcal carriage has hindered vaccine effectiveness^{7,8}.

CHIMs have typically been conducted in healthy, young adult volunteers. These are generally the population least affected by infectious diseases, and so efforts are being made to increase the relevance of CHIM models. For example, the UK EHPC model has been used to study pneumococcal carriage in older adults, which has indeed demonstrated a distinct infectious disease process compared to young adults⁹. The next step for CHIMs in sub-Saharan Africa should also focus on more at-risk populations. Careful consideration of safety, public perception, and justification of the research is crucial. In Malawi, people living with HIV (PLHIV) are a key population to study pneumococcal disease and carriage and an EHPC in PLHIV in Malawi is now ongoing¹⁰.

Our review of the published literature demonstrates no other CHIM in PLHIV, and no framework for CHIMs in PLHIV. Ethical frameworks and considerations for CHIMs in general have been explored, but do not address CHIMs in a potentially vulnerable population^{4,11-14}. CHIMs must be justified by value of the research and must present an acceptable risk both to participants and the wider community. Here we adapt the ethical frameworks developed by Binik (2020) and Miller and Grady (2001), which evaluate CHIMs in the general population, to apply them to a potentially vulnerable population such as PLHIV^{13,14}. We explore the case for undertaking a CHIM in PLHIV using the EHPC in PLHIV in Malawi as a case-example. Additionally, we present feedback from two patient and public involvement and engagement (PPIE) workshops in Malawi in which attendees were invited to share views on CHIM in PLHIV.

A considerations framework for controlled human infection models involving people living with HIV

Our proposed framework consists of four ethical questions, each encompassing several considerations (Table 1).

1. Is the scientific rationale for a controlled human infection model in PLHIV justified, and does the research have translational value for the populations affected?

PLHIV are a key population in addressing the burden of pneumococcal carriage and disease in settings with high HIV prevalence such as Malawi. PLHIV exhibit higher rates of pneumococcal carriage compared to HIV-uninfected and paradoxically exhibit even higher rates once established on antiretroviral therapy^{7,15}. Pneumococcal carriage is a prerequisite for invasive pneumococcal disease (IPD) and despite the positive impact of antiretroviral therapy in reducing risk of IPD,

Ethical question	Considerations
1. Is the scientific rationale justified and does the research have translational value for the populations affected?	 Justification of the research question in the at -risk population Adequacy of the CHIM study design Justification of conducting the study in a low-income setting
2. Are the risks and burden of the research acceptable and can they be minimised?	Risk and burden to individual participantRisk and burden to communityRisk and burden to local infrastructure
3. What are the specific vulnerabilities of the target population, and should this vulnerable group be enrolled?	 Physiological vulnerability Socio-economic vulnerability (e.g. financial hardship) Logistical vulnerability (e.g. frequent medical appointments)
4. What is the perception from local community stakeholders and participants of controlled human infection models in the at-risk population?	Perception of risksPerception of benefitsCultural factors

 Table 1. Framework for planning a controlled human infection model in an at-risk population such as PLHIV.

PLHIV remain at increased risk compared to HIV-uninfected individuals¹⁶. In addition, pneumococcal carriage is a precursor for transmission and PLHIV may be one of the drivers of ongoing transmission observed in Malawi following infant pneumococcal conjugate vaccine (PCV13) immunisation⁷. Pneumococcal carriage is also a precursor for emergence of antimicrobial resistance (AMR) and PLHIV could act as a reservoir of AMR¹⁷.

There is currently no vaccination policy for PLHIV in sub-Saharan Africa. The pneumococcal polysaccharide vaccine (PPV23) was harmful in PLHIV in Uganda^{18,19}. The 7-valent pneumococcal conjugate vaccine (PCV7) was safe and effective against vaccine-serotype disease in PLHIV in Malawi, however has limited serotype coverage and no evidence for effectiveness against carriage in PLHIV²⁰. Further data on vaccine efficacy in PLHIV, immune correlates of protection from pneumococcal carriage, and dynamics of pneumococcal carriage in PLHIV are urgently required for an evidence-based vaccination strategy in sub-Saharan Africa and can be delivered by the EHPC CHIM in PLHIV.

Adequacy of study design: CHIMs allow researchers to control for factors which are often unknown in a community study including participant pre-exposure, microbiological and immunological status; dose, route, and strain of infectious inoculum; and environmental factors which may influence the pathogenesis of infection. Data exist on pneumococcal carriage in PLHIV, and how this differs from HIV-uninfected individuals, yet it is unclear whether this is a result of differences in immunity, or environmental and demographic factors²¹. In addition, CHIMs offer the unique opportunity for cost-effective vaccine evaluations, which require fewer participants and a shorter studyduration than a prospective community-based study. CHIMs can efficiently up- and down-select vaccine candidates, as described on the malaria CHIM²².

Justification of study setting in a low-income country: Conducting CHIM studies requires good clinical services, microbiology, and governance structures to allow them to be conducted safely and to acceptable standards¹². The World Health Organisation (WHO) released guidance highlighting the value of CHIMs for developing infrastructure and research capacity, where it meets local health priorities²³. The setting in Malawi is essential to answer the pertinent questions on pneumococcal carriage and vaccine efficacy in PLHIV in sub-Saharan Africa. Distinct pneumococcal vaccine efficacy has been demonstrated between Malawi and the UK⁸. PLHIV in the UK can be expected to be different from Malawi or other low-income settings in terms of immune profile and vaccine response, as they are exposed to a higher force of infection than their UK counterparts and benefit from herd and direct immunity from PCV²⁴.

2. Are the risks and burden of the research acceptable and can they be minimised?

Participant safety is paramount when recruiting a potentially vulnerable group like PLHIV. This is important for study integrity, and participant and public trust in CHIMs. Binik (2020)

advocates distinguishing study risks and burdens¹³. CHIM studies may demand significant burden on participants, such as isolation and frequent sampling, even if risk of serious harm is minimal¹³. CHIM studies must also consider risks and burdens to the local community and local infrastructure.

Risk and burden to individual participant - Individual risk may be evaluated by striving for a minimal risk study i.e. risk is no greater than those presented in daily life¹³. Pneumococcal carriage is ubiquitous in PLHIV in Malawi, and thus may be considered a risk presented in daily life for this population. At any one time, 26% to 52% of PLHIV will have natural pneumococcal carriage, and 99% of PLHIV will experience at least one pneumococcal carriage event over six to ten months^{7,15,25}. Most of these events do not result in IPD and experimental carriage is expected to be a similar benign event¹⁰. However, it is prudent to consider experimental pneumococcal inoculation more than minimal risk to ensure comprehensive safety monitoring and risk mitigation is incorporated into the study design. This can include isolating or partially isolating participants in study accommodation, regular telephone or in-person contact with participants, careful participant selection, careful inoculum selection and preparation, and comprehensive safety information instruction.

The EHPC in Malawi has recruited only immune-reconstituted and virally-suppressed PLHIV, as risk of IPD closely correlates with CD4 T-cell count^{10,26}. Selection of a relatively benign pneumococcal serotype (6B) which rarely causes disease and employing a slow dose-escalation also mitigates risk²⁷. The EHPC model in Malawi houses inoculated participants in single-occupancy study accommodation and includes frequent telephone contact with participants between study-visits¹⁰. Paradoxically, many safety provisions increase participant burden as study accommodation may take participants away from earning and family responsibilities. Evaluating the burden of the research requires input from the volunteers themselves¹⁴. The EHPC in Liverpool and Malawi has been evaluated in terms of acceptability of sampling and study design to participants, but this must be an ongoing exercise for a CHIM in a vulnerable participant group^{17,28–30}.

Risk and burden to community – Risk of onward transmission exists and is particularly pertinent in PLHIV who exhibit more pneumococcal shedding than HIV-uninfected adults¹⁷. Mitigation of these risks mirror those for participants and includes participant selection to exclude those in contact with vulnerable patient groups, study accommodation to limit community contact, and study clinical care extending to household contacts of participants. However, considering the high natural carriage prevalence in PLHIV, it is unlikely that experimental carriage in PLHIV provides a risk to the community above and beyond natural carriage.

Risk and burden to local infrastructure – Risk to local infrastructure may include burden on health-care facilities with participants attending with inoculum-associated symptoms, and burden on local economy with participants not attending work or unable to fulfil family responsibilities. The EHPC in

3. What are the specific vulnerabilities of the target population and should this vulnerable group be enrolled?

Vulnerabilities in PLHIV are physiological, socioeconomic and logistical. The EHPC in PLHIV in Malawi has been designed to provide PLHIV with population-specific information prior to participation, to account for differences in health-literacy and understanding of infection and disease¹⁴. PLHIV are under regular medical care and avoiding interference with this due to frequent study visits is essential, as well as mitigating against accidental serostatus disclosure in the study design. Although serious illness because of experimental pneumococcal challenge is unlikely, PLHIV may be living with undiagnosed complication which are unveiled during study recruitment and warrant further medical assessment which may not be easily accessible.

Furthermore, PLHIV in Malawi experience more financial hardship than the HIV-uninfected population which could lead to undue influence (by remuneration)²¹. Equally, selective non-recruitment of PLHIV because of financial hardship might be deemed unethical as financial discrimination. Financial reimbursement on the EHPC in Malawi has been determined by local guidelines on compensation, regardless of participants' serostatus or vulnerability³¹. Exclusion on the basis of perceived specific vulnerabilities may be considered paternalistic, and prevents PLHIV exercising their autonomy in the context of fully informed consent¹³. Key to this process is active consultation and engagement with the community advisory groups and the local ethics committees.

4. What is the perception from local community stakeholders and participants of controlled human infection models in the at-risk population?

Public and participant perception is a key consideration for CHIM studies, and community consultation and stakeholder engagement prior to, and during implementation is vital to permit CHIMs to be conducted and for enhancing the scientific and ethical quality of the studies. A step-wise approach to community consultation has been described in Kenya and Malawi^{4,28,30}. PPIE provides a framework for CHIMs to implement safe, ethical, and acceptable studies. For example, the use of passive recruitment techniques in the EHPC in PLHIV was implemented as direct solicitation was identified by stakeholders to potentially risk unduly pressuring individuals to participate^{10,30}. Although this approach also has potential to introduce bias, hence reducing the generalisability of the data.

The perception of CHIM in PLHIV from local community stakeholders and CHIM participants was explored in Malawi in two workshops in October 2022 and January 2023. A total of 37 attendees included PLHIV, healthcare providers, community leaders, community advisory groups, former pneumococcal CHIM participants, and ethical review board members. The perception of CHIM in PLHIV reported by attendees reflect the four ethical questions presented in our framework (Table 1). The scientific rationale or value of the research was framed in terms of reducing the burden of respiratory illness which would "reduce much workload on [the] health sector" and in terms of generating locally relevant data to inform local vaccine programmes:

> "Why do we have to wait for results from studies done elsewhere? This is a welcome development"

Attendees at the workshop focused on individual risk and risk mitigation, rather than community or local infrastructure risk, and focused on risk rather than burden. Attendees largely supported acceptability of CHIM in PLHIV if risk to individual participants was mitigated, for example by recruiting participants who were considered stable and well:

"I agree that this research can be done in people who are infected with HIV but [they] must be stable"

"I will say two ways, yes and no. Yes, if the study will focus on PLHIV with good HIV control and no if the patient is severely sick"

Acceptability was also contingent on mitigating risk with robust safety procedures so "if a person gets sick, they can meet [the research team] easily" and "there should be a plan so that if there is any problem, the person can get help quickly". Attendees highlighted that these safety procedures should be clearly understood by participants:

"My opinion is that these people should be well protected and we should give them a chance to ask questions"

The need to maintain safety even when a participant withdraws consent was mentioned so "*if a person wants to quit they should quit safely*". Although the risk and burden on local infrastructure and community was not directly raised by attendees, robust safety procedures necessary for individual participant safety was recognised as a mitigation against burdening local health infrastructure.

The specific vulnerabilities of PLHIV identified by attendees included differences in health literacy which would influence obtaining informed consent. The need to provide sufficient information to PLHIV participants was identified:

"Let's give full information so that if they want to join the study they know what it involves"

"Sit down with them and enlighten them thoroughly so that they understand"

Finally, the central role of community consultation for a successful CHIM in PLHIV was repeatedly mentioned by

attendees at the PPIE workshops in Malawi. Attendees highlighted the need to prevent unfounded myths and fears spreading in the community:

"If people are not told, they will not understand and we see them as afraid because they have not been taught [...]"

"The study seems safe but there is need to involve the community throughout the study to clear out myths"

The need to be sensitive to potential vaccine hesitancy was raised:

"...it is important is that people are educated well, because we already have challenges in the communities to convince them to have some of these vaccines"

The workshop recommended continuous community engagement prior to, and during the study to clarify myths and build a community strategy for future challenge models.

Conclusions

The case for studying experimental pneumococcal carriage in PLHIV in Malawi is strong as there is a need to address the ongoing burden in PLHIV and the high post-vaccine residual pneumococcal carriage in Malawi. Like any human subject study, there must be a careful consideration of the potential benefits of the study, off set against the potential risks and burdens. Initial stakeholder consultation demonstrates support for CHIMs in PLHIV in Malawi as long as individual risk is mitigated against and community consultation and education is prioritised. The EHPC in PLHIV in Malawi demonstrates that this can be done safely and acceptably however ongoing community engagement and consultation is essential.

List of abbreviations

AMR = antimicrobial resistance

CHIM = Controlled Human Infection Model

EHPC = Experimental Human Pneumococcal Carriage model

IPD = Invasive pneumococcal disease

PCV7 = 7-valent pneumococcal conjugate vaccine

PCV13 = 13-valent pneumococcal conjugate vaccine

PPV23 = pneumococcal polysaccharide vaccine

PLHIV = people living with HIV

PPIE = patient and public involvement and engagement

Ethics and consent statement

Ethical approval was not required to conduct the stakeholder consultation meeting. This was part of community engagement conducted by the Malawi-Liverpool Wellcome programme and the MARVELS consortium. Verbal consent was sought to engage the PPIE participants and record the discussion.

Data availability

No data are associated with this article

Author contributions

KD, AC, SG, KJ, DF conceived the project. AC collected the stakeholder data. KD reviewed the literature and developed the framework. AC, SS, AK, EN, PL curated the data. AC and KD analysed the data and wrote the first draft of the manuscript. SG, KJ, DF, HM supervised the project. All authors reviewed and edited the final manuscript.

Acknowledgements

We are grateful to all attendees at the PPIE workshops in Malawi.

References

- Morton B, Burr S, Chikaonda T, et al.: A feasibility study of controlled human infection with Streptococcus pneumoniae in Malawi. eBioMedicine. 2021; 72: 103579.
 - PubMed Abstract | Publisher Full Text | Free Full Text
- Egesa M, Ssali A, Tumwesige E, et al.: Ethical and practical considerations arising from community consultation on implementing controlled human infection studies using Schistosoma mansoni in Uganda. Glob Bioeth. 2022; 33(1): 78–102.
 - PubMed Abstract | Publisher Full Text | Free Full Text
- Gordon SB, Sichone S, Chirwa AE, et al.: Practical considerations for a TB Controlled Human Infection Model (TB-CHIM); the case for TB-CHIM in Africa, a systematic review of the literature and report of 2 workshop discussions in UK and Malawi [version 2; peer review: 2 approved, 1 approved with reservations]. Wellcome Open Res. 2023; 8: 71. PubMed Abstract | Publisher Full Text | Free Full Text
- Hodgson SH, Juma E, Salim A, et al.: Lessons learnt from the first controlled human malaria infection study conducted in Nairobi, Kenya. Malar J. 2015; 14: 182.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Shekalaghe S, Rutaihwa M, Billingsley PF, et al.: Controlled human malaria infection of Tanzanians by intradermal injection of aseptic, purified,

cryopreserved Plasmodium falciparum sporozoites. Am J Trop Med Hyg. 2014; 91(3): 471–80.

PubMed Abstract | Publisher Full Text | Free Full Text

- Honkpehedji Y, Agobe JCD, Zinsou F, et al.: Plasmodium falciparum parasite dynamics determined by qPCR after controlled human malaria infection in semi-immunes from Gabon. BMJ Glob Health. 2017; 2(Suppl 2). Publisher Full Text
- Swarthout TD, Fronterre C, Lourenço J, et al.: High residual carriage of vaccine-serotype Streptococcus pneumoniae after introduction of pneumococcal conjugate vaccine in Malawi. Nat Commun. 2020; 11(1): 2222. PubMed Abstract | Publisher Full Text | Free Full Text
- Dula D, Morton B, Chikaonda T, et al.: Effect of 13-valent pneumococcal conjugate vaccine on experimental carriage of Streptococcus pneumoniae serotype 6B in Blantyre, Malawi: a randomised controlled trial and controlled human infection study. Lancet Microbe. 2023; 4(9): e683–e691. PubMed Abstract | Publisher Full Text | Free Full Text
- Adler H, German EL, Mitsi E, et al.: Experimental human pneumococcal colonization in older adults Is feasible and safe, not immunogenic. Am J Respir Crit Care Med. 2021; 203(5): 604–13.
 PubMed Abstract | Publisher Full Text | Free Full Text
- 10. Doherty K, Dula D, Chirwa A, et al.: Experimental pneumococcal carriage

in people living with HIV in Malawi: the first controlled human infection model in a key at-risk population [version 1; peer review: 3 approved]. Wellcome Open Res. 2024; 9: 2. PubMed Abstract | Publisher Full Text | Free Full Text

11. Hope T, McMillan J: Challenge studies of human volunteers: ethical issues. J Med Ethics. 2004; 30(1): 110–6.

PubMed Abstract | Publisher Full Text | Free Full Text

- Gordon S, Rylance J, Luck A, et al.: A framework for Controlled Human Infection Model (CHIM) studies in Malawi: Report of a Wellcome Trust workshop on CHIM in Low Income Countries held in Blantyre, Malawi [version 1; peer review: 2 approved]. Wellcome Open Res. 2017; 2: 70. PubMed Abstract | Publisher Full Text | Free Full Text
- Binik A: What risks should be permissible in controlled human infection model studies? *Bioethics.* 2020; 34(4): 420–30.
 PubMed Abstract | Publisher Full Text
- Miller FG, Grady C: The ethical challenge of infection-inducing challenge experiments. *Clin Infect Dis.* 2001; 33(7): 1028–33.
 PubMed Abstract | Publisher Full Text
- Heinsbroek E, Tafatatha T, Phiri A, et al.: Persisting high prevalence of pneumococcal carriage among HIV-infected adults receiving antiretroviral therapy in Malawi: a cohort study. AIDS. 2015; 29(14): 1837–44.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Nunes MC, von Gottberg A, de Gouveia L, et al.: Persistent high burden of invasive pneumococcal disease in South African HIV-infected adults in the era of an antiretroviral treatment program. *PLoS One*. 2011; 6(11): e27929. PubMed Abstract | Publisher Full Text | Free Full Text
- Sibale L, Phiri J, Mitole N, et al.: Frequent shedding of multi-drug resistant pneumococci among adults living with HIV on suppressive antiretroviral therapy in Malawi. medRxiv. 2022; 2022.10.28.22281638.
 Publisher Full Text
- French N, Nakiyingi J, Carpenter LM, et al.: 23-valent pneumococcal polysaccharide vaccine in HIV-1-infected Ugandan adults: double-blind, randomised and placebo controlled trial. Lancet. 2000; 355(9221): 2106–11. PubMed Abstract | Publisher Full Text
- Watera C, Nakiyingi J, Miiro G, et al.: 23-Valent pneumococcal polysaccharide vaccine in HIV-infected Ugandan adults: 6-year follow-up of a clinical trial cohort. AIDS. 2004; 18(8): 1210–3.
 - PubMed Abstract | Publisher Full Text
- French N, Gordon SB, Mwalukomo T, et al.: A trial of 7-valent pneumococcal conjugate vaccine in HIV-infected adults. N Engl J Med. 2010; 362(9): 812–22. PubMed Abstract | Publisher Full Text | Free Full Text
- 21. Thindwa D, Mwalukomo TS, Msefula J, et al.: Risk factors for pneumococcal carriage in adults living with HIV on antiretroviral therapy in the infant

pneumococcal vaccine era in Malawi. *medRxiv*. 2022. Publisher Full Text

- Sauerwein RW, Roestenberg M, Moorthy VS: Experimental human challenge infections can accelerate clinical malaria vaccine development. Nat Rev Immunol. 2011; 11(1): 57–64.
 PubMed Abstract | Publisher Full Text
- WHO: WHO guidance on the ethical conduct of controlled human infection studies. 2021; (accessed September 14, 2023). Reference Source
- Lourenço J, Obolski U, Swarthout TD, et al.: Determinants of high residual post-PCV13 pneumococcal vaccine-type carriage in Blantyre, Malawi: a modelling study. BMC Med. 2019; 17(1): 219.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Carrim M, Tempia S, Thindwa D, et al.: Unmasking pneumococcal carriage in a high HIV prevalence population in two community cohorts with a high prevalence of HIV in South Africa, 2016-2018: the PHIRST study. Clin Infect Dis. 2023; 76(3): e710–e717.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Gilks CF, Ojoo SA, Ojoo JC, et al.: Invasive pneumococcal disease in a cohort of predominantly HIV-1 infected female sex-workers in Nairobi, Kenya. Lancet. 1996; 347(9003): 718–23.
 PubMed Abstract | Publisher Full Text
- Song JY, Nahm MH, Moseley MA: Clinical implications of pneumococcal serotypes: invasive disease potential, clinical presentations, and antibiotic resistance. J Korean Med Sci. 2013; 28(1): 4–15. PubMed Abstract | Publisher Full Text | Free Full Text
- Mtunthama Toto N, Gooding K, Kapumba BM, et al.: "At first, I was very afraid" -a qualitative description of participants' views and experiences in the first human infection study in Malawi [version 2; peer review: 2 approved]. Wellcome Open Res. 2021; 6: 89.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Jochems SP, Piddock K, Rylance J, et al.: Novel analysis of immune cells from nasal microbiopsy demonstrates reliable, reproducible data for Immune Populations, and superior cytokine detection compared to nasal wash. *PLoS One*. 2017; 12(1): e0169805.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Kapumba BM, Jambo K, Rylance J, et al.: Stakeholder views on the acceptability of human infection studies in Malawi. BMC Med Ethics. 2020; 21(1): 14.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Gordon SB, Chinula L, Chilima B, et al.: A Malawi guideline for research study participant remuneration [version 2; peer review: 2 approved]. Wellcome Open Res. 2018; 3: 141.
 PubMed Abstract | Publisher Full Text | Free Full Text